Author's response to reviews

Title: Need for a standardized protocol for stress echocardiography in provoking subaortic gradient in various cardiac conditions

Authors:

Pawel Petkow Dimitrow (dimitrow@mp.pl)
Carlos Cotrim (carlosadcotrim@hotmail.com)
Tsung O Cheng (tcheng@mfa.gwu.edu)

Version: 2 Date: 18 May 2014

Author's response to reviews:

Dear Editor

Please find enclosed corrected version of our manuscript to consider for publication in your Journal. All corrections have been marked by red color. The following changes we propose:

1. More precise description of Lafitte observation and conclusions
2. Additional information concerning table 1 was added.
3. Systematic analysis of 468 publications- expanded table 2 with new information
4. Center performing supine measurement of LVOT has been deleted from table 2.
5. Novel information about normal value of transaortic velocity/gradient has been added.
6. Limitation section was extended by important comments.
7. In Table 2 papers describing own experience (two positions) versus different center (remaining positions ) was indicated.
8. Time-line Figure 1 was added.

Sincerely

PP Dimitrow